Author's response to reviews

Title: Comparison of haematological parameters determined by the Sysmex KX-21N automated haematology analyzer and the manual counts

Authors:

Samuel O Ike (sobiajuluike@gmail.com)
Thomas Nubila (tnubila@yahoo.co.uk)
Ernest O Ukaejiofo (eukaejiiofo@yahoo.com)
Imelda Nubila (immelanubila@yahoo.com)
Elvis N Shu (enshu@yahoo.com)
Ifeyinwa Ezema (ifezema@yahoo.com)

Version: 5 Date: 31 March 2010

Author's response to reviews:

1. Objective has been changed to Background in the abstract.
2. Keywords have been removed from the abstract.
3. The email addresses have been added in the title page.
4. The names of the authors are as in the title page. Some of the authors have no second or middle name, so they do not need the initials.
5. Materials and methods has been changed to Methods.
6. Areas of highlights, especially in the discussion section, have been removed.
7. The authors' contributions have been formatted in a paragraph.
8. The tables have been added to the main manuscript, after the references.
9. The legends to the figures have been put at the end of the figures.